
• Tue Jan 3rd: Minor temperature issues when there was MI10 temp controller problem: 

elog230981 & elog231036

– Thanks to Controls & Ops

• Thu Jan 5th: Replaced MI12-B Sump level sensor: elog231121

– Was not able to do a simple drop in replacement; thinking about better solutions

– Looks like the manufacturer may no longer manufacture this model of transducer

– These transducers are also used for MI12 Well Sumps

– Thanks to ISD

• Thu Jan 5th: Very small E:H866 current change to get Autotune to behave: elog231130

– Tuning last week which led to changes
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https://www-bd.fnal.gov/Elog/?orEntryId=230981
https://www-bd.fnal.gov/Elog/?orEntryId=231036
https://www-bd.fnal.gov/Elog/?orEntryId=231121
https://www-bd.fnal.gov/Elog/?orEntryId=231130

